Author's response to reviews

Title: Patient's dissatisfaction with the public and private laboratory services in conducting HIV related testing in Tanzania

Authors:

Sayoki G Mfinanga (gsmfinanga@yahoo.com)  
Amos M Kahwa (akahwa@hotmail.com)  
Sokoine Kivuyo (sokoinele@yahoo.co.uk)  
Godfather Kimaro (fatherdk2002@yahoo.com)  
Beatrice Mutayoba (beatricemutayoba@yahoo.com)  
Esther Ngadaya (engadaya@yahoo.com)  
Rugola Mtandu (rugorugo@yahoo.com)  
Andrew Kilale (kilale@hotmail.com)  
Kijakazi Mashoto (mashotok@yahoo.com)  
Bernad Ngowi (Bngowi@yahoo.com)  
Senkoro Mbazi (mbazisenkoro@yahoo.co.uk)

Version: 6 Date: 5 August 2008

Author's response to reviews: see over
Patient’s dissatisfaction with the public and private laboratory services in conducting HIV related testing in Tanzania

**General:** Authors are pleased to be informed that BMC Health Services Research has accepted and will publish our manuscript (above) pending formatting requirements. The following changes to the format of our manuscript have been made

1) A native English speaking colleague was contacted and has helped to us to copyedit the paper as recommended
2) The list of authors in the manuscript has been written exactly as they are in the submission system.
3) The page 2 of the document containing a repeat of the title, author list and email addresses was removed as requested.
4) The reference list is formatted in the correct form of the journal
5) The rewording of authors’ contribution was done
6) Tables were included in a position after the reference list